Author's response to reviews

Title: Transdermal fentanyl for the treatment of pain caused by osteoarthritis of the knee or hip, an open, multicentre study

Authors:

Xavier Le Loet (Xavier.Le-Loet@chu-rouen.fr)
Karel Pavelka (pavelka@revma.cz)
Ute Richarz (urichar1@jacch.inj.com)

Version: 3 Date: 7 June 2005

Author’s response to reviews:

Author list has been formatted as requested
Author contributions have been added
References have been put into the correct format
All other requested formatting changes have been made (re your e-mail of 2nd June)